Author’s response to reviews

Title: Nature and Prevalence of Adverse drug reaction of Antiretroviral Medications in Halibet National Referral Hospital: a retrospective study.

Authors:
Lidya Hagos (Liduhg@gmail.com)
Siyoma Fessehaye (siyoma2015@gmail.com)
Indermeet Singh Anand (inderlilly@yahoo.com)

Version: 2 Date: 18 Apr 2019

Author’s response to reviews:

Dear Editor,

Thank you so much for your reply and for the good news that our paper has been accepted for publishing when the necessary changes will be made.

As per your request, we have included the heading of conclusion after the discussion before the last paragraph of the discussion. Furthermore, necessary changes have been made in the section; consent to participate and finally a clean manuscript has been uploaded without any reviewed changes.

If there is anything else or you don't find the changes satisfactory, don't hesitate to contact me.

Sincerely,

Lidya